To Editorial Team

Thank you for your message of 4/1/06, and enclosed comments. The paper has been revised according to the reviewer suggestion. The point risen has been addressed, as you will find in the detailed list enclosed. The changes in the revised manuscript are in red. We hope that the revised version of the manuscript will be now suitable for the publication in the BMC Infectious Disease. Thanking you in advance, I look forward for hearing from you.

Best regards,

Sincerely,

Matteo Bassetti, MD, PhD
Clinica Malattie Infettive
A.O. Ospedaliera Universitaria San Martino di Genova
Largo R.Benzi 10
16132 Genova
Italy
Phone +39 010 555 5132
Fax +39 010 3537680
Email: mattba@tin.it, matteo.bassetti@hsanmartino.it

SPECIFIC POINT

1 According to the reviewer's suggestion we investigate the prevalence of Candida albicans and Candida non albicans among the characteristics of patients. These new data are included in the table 1 and in the text.